

Treatment of interstitial lung disease in systemic sclerosis: guidelines and new clinical trial results

Sindhu R. Johnson^a and Elana J. Bernstein^b

Purpose of review

Interstitial lung disease (ILD) is the leading cause of death in patients with systemic sclerosis-associated interstitial lung disease (SSc-ILD). The American College of Rheumatology (ACR), in conjunction with the American College of Chest Physicians (CHEST), recently published clinical practice guidelines for the treatment of adults with systemic autoimmune rheumatic disease-associated ILD, including SSc-ILD. Herein, we summarize evidence from randomized trials evaluating the safety and efficacy of pharmacologic therapies for the treatment of SSc-ILD.

Recent findings

In this review, we present findings from recent randomized controlled trials in SSc-ILD. The pharmacologic therapies discussed include immunosuppressive medications (mycophenolate, cyclophosphamide, rituximab, and tocilizumab) and antifibrotic medications (nintedanib and pirfenidone).

Summary

Randomized trials provide an evidence base for the SSc-ILD treatment recommendations put forth in the ACR/CHEST Guidelines for the treatment of ILD in people with systemic autoimmune rheumatic diseases. These guidelines will help inform clinical practice and highlight areas in which further research is needed.

Keywords

interstitial lung disease, mycophenolate, rituximab, systemic sclerosis, tocilizumab

Systemic sclerosis (SSc) is a complex systemic autoimmune rheumatic disease characterized by immune activation, vasculopathy and fibrosis. While its cause remains unknown, it is thought to occur in those with genetic susceptibility, and possibly, a second environmental exposure [1]. Interstitial lung disease (ILD) is a leading cause of morbidity and mortality in SSc. Risk factors for the development of ILD include antitopoisomerase I antibodies, male sex and African-American race [2–4]. The American College of Rheumatology (ACR) and the American College of Chest Physicians (CHEST) conditionally recommend screening for ILD in people with SSc with pulmonary function tests (PFTs), which include spirometry, lung volumes and diffusion capacity for carbon monoxide (DLCO), as well as high-resolution computed tomography (CT) chest [5[•]]. Once ILD is diagnosed, the ACR/CHEST ILD guidelines conditionally recommend monitoring with PFTs, high-resolution chest CT and ambulatory desaturation testing [5[•]]. The guidelines suggest monitoring ILD with PFTs every 3-6 months in the first year, and then less frequently once stable. Monitoring ILD with ambulatory desaturation testing is suggested every 3–12 months, while monitoring ILD with highresolution chest CT should occur as needed. Monitoring for ILD progression is best achieved in a shared-care model with pulmonology.

The past decade has seen significant advances in treatment options for SSc-ILD. The ACR/CHEST ILD guidelines for the treatment of ILD in SSc present a menu of options (Fig. 1) that are 'preferred' as well as those that could be considered 'additional' options [6[•]]. While there have been many observational studies evaluating the effectiveness of pharmacologic agents for SSc-ILD, in this article, we summarize the randomized trials evaluating the efficacy

Correspondence to Elana J. Bernstein, MD, MSc, 630 West 168th Street, Suite 3-450, New York, NY 10032, USA. E-mail: ejb2153@cumc.columbia.edu

Curr Opin Rheumatol 2024, 36:420–426 DOI:10.1097/BOR.000000000001049

^aToronto Scleroderma Program, Schroeder Arthritis Institute, Toronto Western Hospital, Mount Sinai Hospital, Department of Medicine, University of Toronto, Toronto, Ontario, Canada and ^bColumbia University Vagelos College of Physicians and Surgeons, Columbia University Irving Medical Center, New York, New York, USA

KEY POINTS

- Randomized clinical trials provided an evidence base that informed the SSc-ILD treatment recommendations in the ACR/CHEST Guidelines for the treatment of ILD in people with systemic autoimmune rheumatic diseases.
- The ACR/CHEST Guidelines for the treatment of ILD in people with systemic autoimmune rheumatic diseases present a menu of options for the treatment of SSc-ILD.
- Mycophenolate was conditionally recommended as a first-line therapy for SSc-ILD.

and safety of pharmacologic therapies in SSc-ILD (Table 1).

MYCOPHENOLATE

Mycophenolate is conditionally recommended as a first-line treatment for SSc-ILD, and for progression of SSc-ILD if the first ILD treatment was not mycophenolate. There have been two randomized trials to support this recommendation.

The Scleroderma Lung Study II (SLS II) was a randomized, double-blind, parallel group trial conducted at 14 medical centers in the United States comparing mycophenolate mofetil (target dose 1500 mg twice daily) for 24 months to oral cyclophosphamide (target dose 2 mg/kg per day) for 12 months followed by placebo for 12 months [7]. The primary endpoint was the change in forced vital capacity as a percentage of the predicted normal value (FVC% predicted) over the course of 24 months. Both groups demonstrated a median improvement in FVC% predicted: 2.19% [95% confidence interval (95% CI) 0.53–3.84] in the mycophenolate mofetil group and 2.88% (95% CI

1.19–4.58) in the cyclophosphamide group, with no significant difference between groups P = 0.24. This trial has been highly influential in the treatment of SSc-ILD; however, its findings are frequently mislabelled and misinterpreted. Consumers of the medical literature frequently mislabel this study as a 'negative' trial and misinterpret these findings to suggest that mycophenolate and cyclophosphamide have the 'same' effect on FVC. This study was designed as a superiority trial (as opposed to a noninferiority trial). As such, the correct interpretation of this study is that the investigators were unable to reject the null hypothesis of no treatment effect and could not demonstrate that mycophenolate is superior to cyclophosphamide [8]. Thus, it is incorrect to conclude that these medications have the same effect. Leukopenia (n=30 versus n=4 patients)P < 0.05) and thrombocytopenia (four versus zero patients, P < 0.05) occurred significantly more frequently in the cyclophosphamide group. There was no difference in proportion of participants with anemia or pneumonia between treatment arms. There were more serious adverse events in the mycophenolate arm (n=42) than in the cyclophosphamide arm (n=36), but more deaths in the cyclophosphamide arm (n = 11) than in the mycophenolate arm (n = 5). While some investigators use these data to justify mycophenolate as safer than cyclophosphamide, others question whether these differences are clinically significant.

Naidu *et al.* [9] conducted a double-blind, randomized, placebo-controlled pilot trial of mycophenolate mofetil 2000 mg twice daily compared to placebo among 41 patients with SSc-ILD. The primary outcome was median change from baseline FVC at 6 months. The FVC decreased by a median of 2.7% (range -21 to 9, P=0.31) in the mycophenolate group and increased by a median of 1% (range -6



FIGURE 1. Pharmacologic options for the treatment of SSc-ILD.

1040-8711 Copyright © 2024 Wolters Kluwer Health, Inc. All rights reserved.

Copyright © 2024 Wolters Kluwer Health, Inc. All rights reserved.

References	Medications and doses ^a	Sample size ^b	Measure of effect
Naidu et al. [9]	Mycophenolate mofetil up to 2000 mg twice daily Placebo	15 Mycophenolate 19 Placebo	FVC decreased by a median of 2.7% (range – 21 to 9; $P=0.307$) in the mycophenolate group and increased by a median of 1% (range – 6 to 10; P=0.222) in the placebo group. The mean absolute difference in FVC change from baseline to 6 months between mycophenolate and placebo groups was 3.1% (95% Cl – 1.0 to 7.3; $P=0.13$).
Tashkin et al. [7]	Mycophenolate mofetil (target dose 1500 mg twice daily) for 24 months Oral cyclophosphamide (target dose 2 mg/kg per day) for 12 months followed by placebo for 12 months	63 Mycophenolate 63 Cyclophosphamide	FVC% improved from baseline to 24 months by 2.19 in the mycophenolate mofetil group (95% CI 0.53–3.84) and 2.88 in the cyclophosphamide group (95% CI 1.19–4.58). The course of the FVC% did not differ significantly between the two treatment groups (P=0.24), indicating that the trial was negative for the primary endpoint.
Tashkin <i>et al.</i> [10]	Oral cyclophosphamide (≤2 mg/kg of body weight per day) Placebo	73 Cyclophosphamide 72 Placebo	Adjusted 12-month FVC% predicted between the cyclophosphamide and placebo groups was 2.53% (95% CI 0.28 to 4.79, P<0.03) favoring cyclophosphamide
Hoyles et al. [12]	IV cyclophosphamide 600 mg/m ² (mean dose 1050 mg) at 4-week intervals for 6 months, followed by oral azathioprine at 2.5 mg/kg/day (maximum 200 mg/day) Placebo	19 Cyclophosphamide + Azathioprine 18 Placebo	The between-group difference adjusted for baseline FVC was 4.19% (P=0.08).
Nadashkevich et al. [13]	Oral cyclophosphamide 2 mg/kg daily Azathioprine (2.5 mg/kg daily	30 Cyclophosphamide 30 Azathioprine	No change in FVC in the cyclophosphamide group: baseline 90.3 versus 92.5% at 6 months, <i>P</i> = not significant Worsening in FVC in the azathioprine group: baseline 91.7 versus 87.3%, <i>P</i> =0.01
Sircar et al. [14]	IV cyclophosphamide 500 mg/m ² monthly for 6 months Rituximab 1000 mg at 0 and 15 days.	30 Cyclophosphamide 30 Rituximab	Percentage mean FVC in the rituximab group improved from 61.3 to 67.5 and declined in the cyclophosphamide group from 59.3 to 58.1 at 6 months (P=0.003).
Maher et al. [16]	Rituximab 1000 mg at weeks 0 and 15 days Cyclophosphamide (600 mg/m ² body surface area every 4 weeks intravenously for 6 months)	49 Rituximab 48 Cyclophosphamide	Between-group difference in FVC at 24 weeks was -40 ml (P=0.49)
Khanna et al. [17]	Tocilizumab 162 mg subcutaneous weekly Placebo	43 Tocilizumab 44 Placebo	Smaller decline in FVC in tocilizumab group (-34 ml) versus placebo group (-171 ml) at 24 weeks (least square mean difference 136 ml (P=0.04))
Khanna <i>et al</i> . [18]	Tocilizumab 162 mg subcutaneous weekly Placebo	104 Tocilizumab 106 Placebo	Least square mean difference in FVC -4.6 in the placebo group and -0.4 in the tocilizumab group (difference 4.2, P=0.0002)
Acharya et al. [21]	Pirfenidone 2400 mg a day Placebo	17 Pirfenidone 17 Placebo	Stabilization/improvement in FVC occurred in 94.1% pirfenidone group versus 76.5% placebo group (P=0.33)

Tab	le	1.	Summarv	of rand	domized	l trials	evaluat	ina p	harmacol	oaic	treatments	in	SSc-	ILD
			/											

^aPlease note that as of this writing, the only FDA-approved medications for the treatment of SSc-ILD are tocilizumab and nintedanib. All other medications discussed in this manuscript are off label for the treatment of SSc-ILD.

^bNumbers of participants analyzed for the primary outcome. The trials report higher numbers of subjects that were recruited or randomized, but the number of individuals analyzed for the primary outcome is small due to withdrawal or loss to follow-up.

to 10, P=0.22) in the placebo group. The mean absolute difference in FVC change from baseline to 6 months between groups was 3.1% (P=0.13) and not statistically significant. The inability to demonstrate a beneficial effect of mycophenolate over placebo on FVC may include the small sample size, that 6 months of follow-up was too short to detect a treatment effect or that the maximum dose of mycophenolate (3 g/day) was not used. Despite these potential limitations, it is notable that the magnitude of the treatment effect in this trial was comparable to that observed in SLS II. There was no significant difference in the frequency of adverse events between the mycophenolate and placebo groups.

The combined trial evidence suggests that mycophenolate confers a modest beneficial effect on FVC, or at least, results in stabilization of FVC. Moreover, mycophenolate has a more favorable adverse effect profile relative to other options. Furthermore, rheumatologists and pulmonologists have considerable experience with mycophenolate and mycophenolate has a relatively lower cost compared to other options. The combination of evidence, favorable adverse effect profile, experience and cost led to mycophenolate being conditionally recommended for the treatment of SSc-ILD.

CYCLOPHOSPHAMIDE

The Scleroderma Lung Study I (SLS I) was a doubleblind, randomized, placebo-controlled trial comparing oral cyclophosphamide ($\leq 2 \text{ mg}$ per kilogram per day) to placebo, conducted at 13 US centers [10]. The primary outcome was the FVC% predicted at 12 months, after adjustment for the baseline FVC. Seventy-three participants in the cyclophosphamide group and 72 participants in the placebo group were evaluated for the primary outcome. The mean absolute difference in adjusted 12-month FVC% predicted between the cyclophosphamide and placebo groups was 2.53% (95% CI 0.28–4.79, P < 0.03) favoring cyclophosphamide. Oral cyclophosphamide resulted in a modest beneficial effect or stabilization of FVC at 12 months but the effects on FVC collapsed by 24 months [11]. There was a greater frequency of adverse events in the cyclophosphamide group, but the difference in serious adverse events between the two groups was not significant. Leukopenia (n = 19 versus n = 0 patients, P < 0.05) and neutropenia (n=7 patients versus n=0patients, P < 0.05) were more common in the cyclophosphamide group than in the placebo group, with no statistically significant differences in hematuria, anemia or pneumonia between groups.

Hoyles *et al.* [12] reported a double-blind, randomized, placebo-controlled trial comparing

intravenous cyclophosphamide 600 mg/m^2 (mean dose 1050 mg) every 4 weeks for 6 months, followed by oral azathioprine 2.5 mg/kg/day (maximum 200 mg/day) to placebo. The cyclophosphamide group also received prednisolone 20 mg orally on alternate days. Nineteen patients received cyclophosphamide followed by azathioprine, and 18 patients received placebo for 1 year. The primary outcome measures were change in FVC% predicted and change in DLCO. The between-group difference adjusted for baseline FVC was 4.19% (P=0.08). Although not statistically significant, the magnitude of this modest effect on FVC is comparable to that observed in SLS I. No bone marrow suppression or hemorrhagic cystitis was observed.

Nadashkevich *et al.* [13] reported a randomized, unblinded, trial conducted in Canada and the Ukraine, comparing oral cyclophosphamide (2 mg/kg daily for 12 months followed by a maintenance dose of 1 mg/kg daily) to azathioprine (2.5 mg/kg daily for 12 months followed by a maintenance dose of 2 mg/kg daily). There was no significant change in FVC% predicted in the cyclophosphamide group (baseline 90.3 versus 92.5% at 6 months, P = not significant), whereas there was a statistically significant worsening in FVC% predicted in the azathioprine group (baseline 91.7 versus 87.3%, P = 0.01).

RITUXIMAB

Sircar *et al.* [14] conducted a randomized trial comparing monthly intravenous cyclophosphamide 500 mg/m^2 for 24 weeks to two doses of rituximab 1000 mg at 0 and 15 days. The primary outcome was FVC% predicted at 6 months. The mean FVC% predicted in the rituximab group improved from 61.3 to 67.5% and declined in the cyclophosphamide group from 59.3 to 58.1% at 6 months (*P*=0.003). Serious adverse events were more common in the cyclophosphamide group.

The DESIRES trial was a double-blind, randomized, placebo-controlled trial conducted at four centers in Japan comparing rituximab (375 mg/m²) weekly for 4 weeks to placebo [15]. The primary endpoint was the absolute change in mRSS at 24 weeks. One of the secondary endpoints was absolute change in FVC% predicted at 24 weeks. Twentyseven patients in the rituximab group and 22 patients in the placebo group received at least one dose and completed 24 weeks of follow-up. The absolute change in mRSS was lower in the rituximab group than in the placebo group (-6.3 versus 2.14, P < 0.0001). The absolute change in FVC% predicted was 0.09% in the rituximab group and -2.87% in the placebo group (P=0.044). Adverse events were similar in both groups. The most common adverse event was upper respiratory infection (n = 11 ritux-imab group, n = 10 placebo group).

The RECITAL trial was a randomized, doubleblind, double-dummy, controlled trial comparing rituximab (1000 mg at weeks 0 and 2 intravenously) to cyclophosphamide $(600 \text{ mg/m}^2 \text{ body surface area})$ every 4 weeks intravenously for 6 months) in patients with severe or progressive ILD related to SSc, idiopathic inflammatory myositis or mixed connective tissue disease conducted at 11 centers in the United Kingdom [16]. The primary endpoint was change in FVC at 24 weeks compared with baseline. Forty-eight participants in the cyclophosphamide group and 49 in the rituximab group received at least one dose of treatment. At 24 weeks, FVC was improved from baseline in both the cyclophosphamide group (unadjusted mean increase 99 ml) and the rituximab group (97 ml). The difference in the primary endpoint at 24 weeks was -40 ml (P = 0.49) between the rituximab and cyclophosphamide groups. Gastrointestinal and respiratory issues were the most commonly reported adverse events in both groups. This trial was unable to demonstrate that rituximab is superior to cyclophosphamide, as both treatments resulted in increased FVC.

TOCILIZUMAB

The faSScinate study was a multicenter, international, randomized, double-blind, placebo-controlled phase 2 trial comparing subcutaneous tocilizumab 162 mg weekly to placebo [17]. The primary endpoint was the difference in mean change from baseline in mRSS at 24 weeks. Change in FVC at 48 weeks was an exploratory endpoint. Forty-three patients were assigned to tocilizumab and 44 assigned to placebo. The least squares mean change in mRSS at 24 weeks was -3.92 in the tocilizumab group and -1.22 in the placebo group (difference -2.70, P = 0.09). Fewer patients in the tocilizumab group than in the placebo group had a decline in FVC% predicted at 48 weeks (P = 0.04). Serious infections were more common in the tocilizumab group (n = 7 [16%]) than in the placebo group (n = 2 [5%]).

The focuSSced trial was a multicenter, international, randomized, double-blind, placebocontrolled phase 3 trial comparing tocilizumab 162 mg subcutaneous weekly to placebo for 48 weeks [18]. The primary outcome was the difference in change in mRSS from baseline to week 48. FVC% predicted at week 48 was a secondary outcome. The adjusted difference in mRSS was -1.73 (P = 0.10). The least square mean change from baseline in FVC% predicted was -4.6 in the placebo group and -0.4 in the tocilizumab group with a difference of 4.2 (P = 0.0002). There were 54 (52%) adverse events in the tocilizumab group and 53 (50%) in the placebo group.

The combined data suggest that tocilizumab has a modest beneficial effect or preserves lung function in people with early SSc-ILD, specifically those with diffuse cutaneous SSc and elevated acute-phase reactants. These data led to the approval of tocilizumab for SSc-ILD by the US Food and Drug administration.

NINTEDANIB

The largest study of SSc-ILD to date – the SENSCIS trial – was a multicenter, international, randomized, placebo-controlled trial [19]. Five hundred seventysix SSc-ILD patients were randomized to receive nintedanib 150 mg twice daily or placebo [20]. About 48.4% of patients had been treated with mycophenolate for at least 6 months before enrollment. The adjusted annual rate of change in FVC over 52 weeks was -52.4 ml/year in the nintedanib group compared to -93.3 ml/year in the placebo group (difference: 40.9 ml/year; P = 0.03) [20]. Nintedanib, in combination with mycophenolate, provided greater numerical preservation of lung function than mycophenolate in combination with placebo, but this was not statistically significant (P=0.45) and SENSCIS was not powered to detect differences in the primary outcome based on baseline mycophenolate use. Diarrhea was reported in 76% (n = 106) of the nintedanib group and 34% (n=48) of the placebo group among those taking mycophenolate at baseline. More data are needed regarding the benefits of initial combination therapy versus a sequential approach in the treatment of SSc-ILD.

PIRFENIDONE

Acharya *et al.* [21] report a double-blind, randomized, placebo-controlled, pilot study comparing pirfenidone 2400 mg per day to placebo for 6 months. The primary outcome was the proportion of subjects with stabilization or improvement in FVC at 6 months. Thirty-four individuals were enrolled. Stabilization/improvement in FVC occurred in 16 (94.1%) of the pirfenidone group and 13 (76.5%) of the placebo group (P=0.33). The median absolute change in FVC% predicted was -0.55 in the pirfenidone group and 1.0 in the placebo group (P=0.51). Common adverse events were gastrointestinal issues and skin rash. This pilot study was unable to demonstrate a beneficial effect of pirfenidone over placebo in improving/stabilizing FVC.

The RELIEF trial was a multicenter, double-blind, randomized, placebo-controlled trial conducted at 17

centers in Germany. One-hundred twenty-seven patients with progressive fibrotic ILD eight 8 (6%) with SSc-ILD were randomized to pirfenidone (titrated up to 801 mg three times per day) or matched placebo, in addition to background therapy [22]. The primary endpoint was change in FVC% predicted from baseline to week 48. The trial was prematurely terminated due to an interim analysis for futility prompted by slow recruitment. Analysis of the available data suggested treatment with pirfenidone resulted in a significantly smaller decline from baseline to week 48 in FVC% predicted than placebo (P = 0.04). Analysis of the SSc-ILD patients was not conducted due to the small sample size.

The Scleroderma Lung Study III was a multicenter, double-blind, placebo-controlled, phase 2 trial evaluating combination mycophenolate mofetil and pirfenidone compared to mycophenolate mofetil and placebo in SSc-ILD [23]. The primary endpoint was the change from baseline in FVC% predicted over 18 months. The investigators were unable to recruit to the prespecified sample size of 150 individuals due to the COVID-19 pandemic and the impact of prior medication use on eligibility. Twenty-seven participants were randomized to mycophenolate with pirfenidone and 24 were randomized to mycophenolate with placebo. Over 18 months, there was a 2.24% improvement in the mycophenolate mofetil with placebo group and a 2.09% improvement in the mycophenolate mofetil with pirfenidone group, and no statistically significant difference between groups (P = 0.93). About 74.1% of individuals on mycophenolate with pirfenidone compared to 29.2% on mycophenolate with placebo had adverse events of special interest (gastrointestinal: 55.6 versus 20.8%, photosensitivity: 14.8 versus 0%).

In summary, clinical trials in SSc-ILD provide an evidence base to inform the ACR/CHEST guidelines for the treatment of SSc-ILD. There are several treatments that provide a modest beneficial effect or stabilize FVC. The choice of pharmacologic option will necessitate shared decision making with the individual patient based on values, preferences, cost and extra-pulmonary manifestation.

Acknowledgements

None.

Financial support and sponsorship

Dr Bernstein's work is supported by K23-AR075112 from NIAMS, R01-HL164758 from NHLBI and W81XWH2210163 from the Department of Defense.

Conflicts of interest

Dr Johnson has no conflicts of interest to declare. Dr Bernstein reports consulting fees from Boehringer Ingelheim and Cabaletta and grant support from Boehringer Ingelheim and aTyr.

REFERENCES AND RECOMMENDED READING

Papers of particular interest, published within the annual period of review, have been highlighted as:

- of special interest
- of outstanding interest
- 1. Alahmari H, Ahmad Z, Johnson SR. Environmental risks for systemic sclerosis. Rheum Dis Clin North Am 2022; 48:845–860.
- Distler O, Assassi S, Cottin V, *et al.* Predictors of progression in systemic sclerosis patients with interstitial lung disease. Eur Respir J 2020; 55: 1902026.
- Hussein H, Lee P, Chau C, Johnson SR. The effect of male sex on survival in systemic sclerosis. J Rheumatol 2014; 41:2193–2200.
- Al-Sheikh H, Ahmad Z, Johnson SR. Ethnic variations in systemic sclerosis disease manifestations, internal organ involvement, and mortality. J Rheumatol 2019; 46:1103–1108.
- Johnson SR, Bernstein EJ, Bolster MB, et al. 2023 American College of
 Rheumatology (ACR)/American College of Chest Physicians (CHEST) Guideline for the Screening and Monitoring of Interstitial Lung Disease in People with Systemic Autoimmune Rheumatic Diseases. Arthritis Care Res

(Hoboken) 2024; 76:1070–1082. This is the first collaboration between the American College of Rheumatology and the American College of Chest Physicians to provide guidelines for the screening and monitoring of SSc-ILD.

- 6. Johnson SR, Bernstein EJ, Bolster MB, et al. 2023 American College of Rheumatology (ACR)/American College of Chest Physicians (CHEST)
- Guideline for the Treatment of Interstitial Lung Disease in People with Systemic Autoimmune Rheumatic Diseases. Arthritis Rheumatol 2024; 76:1182–1200.

This is the first collaboration between the American College of Rheumatology and the American College of Chest Physicians to provide guidelines for the treatment of SSc-ILD.

- Tashkin DP, Roth MD, Clements PJ, et al. Mycophenolate mofetil versus oral cyclophosphamide in scleroderma-related interstitial lung disease (SLS II): a randomised controlled, double-blind, parallel group trial. Lancet Respir Med 2016; 4:708–719.
- Johnson SR, Feldman BM, Pope JE, Tomlinson GA. Shifting our thinking about uncommon disease trials: the case of methotrexate in scleroderma. J Rheumatol 2009; 36:323–329.
- Naidu G, Sharma SK, Adarsh MB, et al. Effect of mycophenolate mofetil (MMF) on systemic sclerosis-related interstitial lung disease with mildly impaired lung function: a double-blind, placebo-controlled, randomized trial. Rheumatol Int 2020; 40:207–216.
- Tashkin DP, Elashoff R, Clements PJ, et al. Cyclophosphamide versus placebo in scleroderma lung disease. N Engl J Med 2006; 354:2655–2666.
- Tashkin DP, Elashoff R, Clements PJ, et al. Effects of 1-year treatment with cyclophosphamide on outcomes at 2 years in scleroderma lung disease. Am J Respir Crit Care Med 2007; 176:1026–1034.
- 12. Hoyles RK, Ellis RW, Wellsbury J, et al. A multicenter, prospective, randomized, double-blind, placebo-controlled trial of corticosteroids and intravenous cyclophosphamide followed by oral azathioprine for the treatment of pulmonary fibrosis in scleroderma. Arthritis Rheum 2006; 54:3962–3970.
- Nadashkevich O, Davis P, Fritzler M, Kovalenko W. A randomized unblinded trial of cyclophosphamide versus azathioprine in the treatment of systemic sclerosis. Clin Rheumatol 2006; 25:205–212.
- Sircar G, Goswami RP, Sircar D, et al. Intravenous cyclophosphamide vs rituximab for the treatment of early diffuse scleroderma lung disease: open label, randomized, controlled trial. Rheumatology (Oxford) 2018; 57:2106– 2113.
- Ebata S, Yoshizaki A, Oba K, et al. Safety and efficacy of rituximab in systemic sclerosis (DESIRES): a double-blind, investigator-initiated, randomised, placebo-controlled trial. Lancet Rheumatol 2021; 3:e489–e497.
- 16. Maher TM, Tudor VA, Saunders P, et al. Rituximab versus intravenous cyclophosphamide in patients with connective tissue disease-associated interstitial lung disease in the UK (RECITAL): a double-blind, double-dummy, randomised, controlled, phase 2b trial. Lancet Respir Med 2023; 11:45–54.
- Khanna D, Denton CP, Jahreis A, *et al.* Safety and efficacy of subcutaneous tocilizumab in adults with systemic sclerosis (faSScinate): a phase 2, randomised, controlled trial. Lancet 2016; 387:2630–2640.

- Khanna D, Lin CJF, Furst DE, et al. Tocilizumab in systemic sclerosis: a randomised, double-blind, placebo-controlled, phase 3 trial. Lancet Respir Med 2020; 8:963–974.
- Assassi S, Distler O, Allanore Y, et al. Effect of nintedanib on progression of systemic sclerosis-associated interstitial lung disease over 100 weeks: data from a randomized controlled trial. ACR Open Rheumatol 2022; 4:837–844.
- Distler O, Highland KB, Gahlemann M, et al. Nintedanib for systemic sclerosis-associated interstitial lung disease. N Engl J Med 2019; 380: 2518–2528.
- Acharya N, Sharma SK, Mishra D, et al. Efficacy and safety of pirfenidone in systemic sclerosis-related interstitial lung disease-a randomised controlled trial. Rheumatol Int 2020; 40:703–710.
- 22. Behr J, Prasse A, Kreuter M, et al. Pirfenidone in patients with progressive fibrotic interstitial lung diseases other than idiopathic pulmonary fibrosis (RELIEF): a double-blind, randomised, placebo-controlled, phase 2b trial. Lancet Respir Med 2021; 9:476–486.
- Khanna D, Spino C, Bernstein EJ, et al. Combination therapy of mycophenolate mofetil and pirfenidone vs. mycophenolate alone: results from the Scleroderma Lung Study III [abstract]. Arthritis Rheumatol 2022; 74(suppl 9).